Department of Mathematics and Computer Science University of Southern Denmark, Odense

February 18, 2005 LMF DM69 — Lecture 4

Lecture 4 — February 23

• Sections 3.6–3.7 in Bang-Jensen and Gutin.

Problems for February 25

1. Problems 3.34–3.35 in Bang-Jensen and Gutin. 2. Problem 26-3 in Cormen.

3. Problem 3.30 in Bang-Jensen and Gutin. 4. Explain Section 3.8 in Bang-Jensen and Gutin. 5. Explain Section 3.9 in Bang-Jensen and Gutin. 6. Problem 26.2-9 in Cormen (note that the graph is undirected).